Author’s response to reviews

Title: Single coronary ostium in a patient with quadricuspid aortic valve combined with aneurysmal ascending aortic dilatation

Authors:

Do Yeon Kim (goodmd77@hanmail.net)

Hwan Wook Kim (kimhwanwook@catholic.ac.kr)

Version: 1 Date: 16 Jun 2017

Author’s response to reviews:

Response to reviewers

Reviewer #1: Dear All,

I read your article with great interest.

My only comment is that - based on recent publications - QAV is associated with aortic dilatation in 30% of the cases.

Excellent work

Congratulations.

Answer: Thanks for your comments. We really appreciate your effort in reviewing my manuscript.

Reviewer #2: Many congratulations to the authors of this article. The content of the article and the images are indeed very informative. I would advise some improvement of the written English of the text of the article in particular the first paragraph of the background. LV and LM should appear in the list of abbreviations. Best of luck!

Answer: Thanks for your comments. We revised the manuscript with the assistance of native English person.

Changes in the first paragraph -

1) ~ reported only 2 cases in 6000 autopsies ~
-> reported in only 2 cases of 6000 autopsies ~

2) enable us to know the precise anatomy with configurations of this morphology together with those of other accompanying ~

-> enable us to reveal precise anatomical configurations of this morphology along with other accompanying ~

Answer: We added LV, LM and RCA in the list of abbreviations.